Author’s response to reviews

Title: An Immunocompetent Patient with a Nonsense Mutation in NHEJ1 Gene

Authors:

Hossein Esmaeilzadeh (esmailzadeh ho@yahoo.com)
Mohammad Reza Bordbar (mbordbar53@gmail.com)
Zahra Hojaji (zahra_hojaji@yahoo.com)
Parham Habibzadeh (parham.habibzadeh@yahoo.com)
Dorna Afshinfar (Afshinfar.dorna@gmail.com)
Mohammad Miryounesi (miryounesi@gmail.com)
Majid Fardaei (fardaei.majid@gmail.com)
Mohammad Ali Faghihi (mfaghihi@med.miami.edu)

Version: 2 Date: 10 Mar 2019

Author’s response to reviews:

1. Please provide the heading "Declarations" for the corresponding section.
   Done

2. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
   Done

3. Please rename:
   - 'Ethics Statement' heading to 'Ethics approval and consent to participate'
   - 'Conflicts of Interest Statement' heading to 'Competing Interest'
   Done

4. Thank you for providing your response to reviewers, CARE checklist and cover letter. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.
   The mentioned parts were removed.
5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done